Author’s response to reviews

Title: The effect of Liv-52 on liver ischemia reperfusion damage in rats

Authors:
Orhan Cimen (orhancimen24@hotmail.com)
Huseyin Eken (huseyineken80@hotmail.com)
Ferda Keskin Cİmen (ferdakeskinn@hotmail.com)
Arif Burak Cekic (abceekic@hotmail.com)
Nezahat Kurt (nezahatkurt@hotmail.com)
Asli Ozbek Bilgin (asliozbekbilgin@gmail.com)
Bahadir Suleyman (bahadirsuleyman@yandex.com)
Halis Suleyman (halis.suleyman@gmail.com)
Renad Mammadov (renad_mamedov@hotmail.com)
Kamil Pehlivanoğlu (pehlivan550@gmail.com)
Eray Kurnaz (erakurnaz@hotmail.com)

Version: 5 Date: 20 Dec 2019

Author’s response to reviews:

Dear editor,

The manuscript revised as you commended.

Sincerely..

Editor Comments:

1. Textual overlap
   We note that the current submission contains some textual overlap with other previously published works, in particular:
   “http://www.scielo.br/scielo.php?lng=en&pid=S0102-
This overlap mainly exists in the Methods.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please rephrase these sections to minimise overlap.

Response: The methods section re-phrased as much as possible.

2. Authors’ contributions

Please note that based on the current authors’ contributions section, ABC, NK, AOB, RM do not automatically qualify for authorship. Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group, alone, does not usually justify authorship.

We ask that you ensure that author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below:

A. Made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;
B. Been involved in drafting the manuscript or revising it critically for important intellectual content;
C. Given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and
D. Agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Please provide more information on individual author contributions and ensure author contributions listed in the authors’ contribution section are reflective of ICMJE guidelines.

Response: All listed authors performed all four points in line with the ICMJE guidelines as it specified and added to the text. All authors have participated sufficiently in the work to take responsibility for appropriate portions of the content; and agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved have read and approved the manuscript. Also given final approval of the version to be published.

3. Availability of data and materials

Please amend your Availability of data and materials statement to ensure it is in line with journal requirements. For you information, data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

- The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

- All data generated or analysed during this study are included in this published article [and its supplementary information files].

Response: All data generated or analysed during this study are included in this published article [and its supplementary information files].

4. Clean copy

Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only. Please remove the cover letter from the file inventory as it is no longer
required at this stage.
Response: It’s done as you commended.